Author’s response to reviews

Title: Trends in childhood and adolescent internalizing symptoms: results from Swedish population based twin cohorts

Authors:

Karolina Sorman (Karolina.Sorman@ki.se)
Natalie Durbeej (Natalie.Durbeej@ki.se)
Eva Norén-Selinus (Eva.Noren@ki.se)
Sebastian Lundström (sebastian.lundstrom@gnc.gu.se)
Paul Lichtenstein (Paul.Lichtenstein@ki.se)
Clara Hellner (Clara.Gumpert@ki.se)
Linda Halldner (Linda.Halldner@ki.se)

Version: 3 Date: 28 Jun 2019

Author’s response to reviews:

Two corrections have been made in this round marked in yellow in the manuscript:

1) The sentence in the abstract has been revised according to the Editor's suggestion
2) A small correction has been made to the family name of one of the authors

AND figures have now been added